Earlier this morning the right Q2 magnet tripped and then tripped again after a hard reset at the power supply in the hall. At the moment, the tech staff have been unable to reset the power supply and are currently investigating the problem. We will likely be down for longer than scheduled. More updates later.


Mark and company detected, located and repaired a broken wire in one of the lead voltage taps on right Q2. The wire was located on a covered connector on the top of the top hat on the magnet. 

This was a factory supplied cable that may have been damaged due to ice buildup on the top hat over time. 

magnet is now operational 

